Course Information Sheet, CMPUT272, Winter 2007

This sheet contains basic information you need to know about this course. For more details see the course web page:

http://ugweb.cs.ualberta.ca/~c272/W07/

Note: All announcements, assignments, and lecture notes will be posted on this web page. It is your responsibility to check this web page frequently (at least twice a week). Instructor Information:

• Instructor: Mohammad R. Salavatipour • E-mail: mreza@cs.ualberta.ca • Office: Ath303 • Phone: 492-1759

The best way to contact me is by e-mail. I usually reply the same day, but if you send me an e-mail on Friday or weekend, don’t expect to get a quick response.

Time tables:

• Lectures:

– Section B1/EB1: Mon,Wed,Fri 10:00-10:50, CEB321. – Section B2/EB2: Tue,Thr 12:30-1:50, CME346.

• Tutorials (seminars):

– Mon, 4:00-4:50, NRE2 001 – Thr, 3:00-3:50, CME 344 • Office hours: Mon 2-3, Tue 3-4, or by appointment.

Text book:

The text book for this course is: ”Discrete Mathematics with Applications”, 3rd Edition, by Susanna S. Epp, Brooks/Cole, 2004.

Grading Scheme:

• Six assignments: 30% (at 5% each).

• Two term tests in class (50 minutes each) 30% (at 15% each). • Final exam: 40%.

Continued on the back ...

Important Dates:

Date Event Mon, Jan 8 Assignment 1 is posted Fri, Jan 26 Assignment 1 due (Assignment 2 is posted) Fri, Feb 9 Assignment 2 due (Assignment 3 is posted) Thu, Feb 15 Test 1 for Section B2/EB2 (Coverage weeks 1-5) Fri, Feb 16 Test 1 for Section B1/EB1 (Coverage weeks 1-5) Feb 19-23 Reading week Fri, Mar 2 Assignment 3 due (Assignment 4 is posted) Fri, Mar 16 Assignment 4 due (Assignment 5 is posted) Thu, Mar 22 Test 2 for Section B2/EB2 (Coverage weeks 6-9) Fri, Mar 23 Test 2 for Section B1/EB1 (Coverage weeks 6-9) Fri, Mar 30 Assignment 5 due (Assignment 6 is posted) Fri, Apr 13 Assignment 6 due Assignments and late policy: All assignments are due at the beginning of the lecture on the due date. Late assignments (submitted up until the beginning of the next lecture on the following Monday) will be deducted 50% of the full mark for that assignment.

Deferred exam (if applicable):

Will be on May 1st, 2007, 9:00-12:00.

Course topics and schedule

week topics Week 1 Chapter 1.1-1.2 Week 2 Chapter 1.3-1.5 Week 3 Chapter 2.1-2.4 Week 4 Chapter 3.1-3.5 Week 5 Chapter 36-3.7, 4.1-4.2 Week 6 Chapter 4.3-4.4 Week 7 Chapter 5.1-5.4 Week 8 Chapter 6.1-6.5 Week 9 Chapter 6.6-6.7, 7.1-7.2 Week 10 Chapter 7.3-7.4 Week 11 Chapter 8.1-8.4 Week 12 Chapter 10.1-10.3 Week 13 Chapter 10.4-10.5

Assignments, policies, and other things:

Check out the information on the webpage of the course.